# TeMA

print ISSN 1970-9889 e-ISSN 1970-9870 FedOA press - University of Naples Federico II Journal of Land Use, Mobility and Environment

DOAJ Rivista scientifica di classe A - 08/F1

Scopus WE

WEB OF SCIENCE



Multilevel scientific approach to impacts of global warming on urban areas, energy transition, optimisation of land use and emergency scenario

Vol.18 n.1 April 2025

TeMA Journal was established with the primary objective of fostering and strengthening the integration between urban transformation studies and those focused on mobility governance, in all their aspects, with a view to environmental sustainability. The three issues of the 2025 volume of TeMA Journal propose articles that deal with the effects of Global warming, reduction of energy consumption, immigration flows, optimization of land use, analysis and evaluation of civil protection plans in areas especially vulnerable to natural disasters and multilevel governance approach to adaptation.

TeMA is the Journal of Land Use, Mobility and Environment and offers papers with a unified approach to planning, mobility and environmental sustainability. With ANVUR resolution of April 2020, TeMA journal and the articles published from 2016 are included in the A category of scientific journals. The articles are included in main scientific database as Scopus (from 2023), Web of Science (from 2015) and the Directory of Open Access Journals (DOAJ). It is included in Sparc Europe Seal of Open Access Journals, and the Directory of Open Access Journals.

# TEMA Journal of Land Use, Mobility and Environment

### NEW CHALLENGES FOR XXI CENTURY CITIES:

Multilevel scientific approach to impacts of global warming on urban areas, energy transition, optimisation of land use and emergency scenario

### 1 (2025)

#### Published by

Laboratory of Land Use Mobility and Environment DICEA - Department of Civil, Architectural and Environmental Engineering University of Naples "Federico II"

TeMA is realized by CAB - Center for Libraries at "Federico II" University of Naples using Open Journal System

Editor-in-Chief: Rocco Papa print ISSN 1970-9889 | online ISSN 1970-9870 Licence: Cancelleria del Tribunale di Napoli, n°6 of 29/01/2008

#### Editorial correspondence

Laboratory of Land Use, Mobility and Environment DICEA - Department of Civil, Building and Environmental Engineering University of Naples "Federico II" Piazzale Tecchio, 80 80125 Naples (Italy)

https://serena.sharepress.it/index.php/tema e-mail: redazione.tema@unina.it

The cover image shows a composition of two photos of the Temple of Serapis in Pozzuoli (Italy). Giuseppe Mazzeo took them in January 2009 and March 2025. At the top, the 2009 image shows the temple flooded, with the pavement not visible. In the down, the 2025 image shows the temple's pavement dry and exposed. The Temple of Serapis is one of the leading visual indicators of the bradyseism phenomenon in the Phlegraean Fields. The bradyseism phase, highlighted by comparison, started in the first years of this century, as shown by the data published by the National Institute of Geophysics and Volcanology (INGV) on the website dedicated to the phenomena (https://www.ov.ingv.it/index.php/il-bradisismo).

TeMA - Journal of Land Use, Mobility and Environment offers researches, applications and contributions with a unified approach to planning and mobility and publishes original inter-disciplinary papers on the interaction of transport, land use and environment. Domains include: engineering, planning, modeling, behavior, economics, geography, regional science, sociology, architecture and design, network science and complex systems.

With ANVUR resolution of April 2020, TeMA Journal and the articles published from 2016 are included in A category of scientific journals. The articles published on TeMA are included in main international scientific database as Scopus (from 2023), Web of Science (from 2015) and the *Directory of Open Access Journals* (DOAJ). TeMA Journal has also received the *Sparc Europe Seal* for Open Access Journals released by *Scholarly Publishing and Academic Resources Coalition* (SPARC Europe). TeMA is published under a Creative Commons Attribution 4.0 License and is blind peer reviewed at least by two referees selected among high-profile scientists. TeMA has been published since 2007 and is indexed in the main bibliographical databases and it is present in the catalogues of hundreds of academic and research libraries worldwide.

#### EDITOR-IN-CHIEF

Rocco Papa, University of Naples Federico II, Italy

#### EDITORIAL ADVISORY BOARD

Mir Ali, University of Illinois, USA Luca Bertolini, University of Amsterdam, Netherlands Luuk Boelens, Ghent University, Belgium Dino Borri, Politecnico di Bari, Italy Enrique Calderon, Technical University of Madrid, Spain Pierluigi Coppola, Politecnico di Milano, Italy Derrick De Kerckhove, University of Toronto, Canada Mark Deakin, Edinburgh Napier University, Scotland Romano Fistola, University of Naples Federico II, Italy Carmela Gargiulo, University of Naples Federico II, Italy Aharon Kellerman, University of Haifa, Israel Nicos Komninos, Aristotle University of Thessaloniki, Greece David Matthew Levinson, University of Minnesota, USA Paolo Malanima, Magna Græcia University of Catanzaro, Italy Agostino Nuzzolo, Tor Vergata University of Rome, Italy Rocco Papa, University of Naples Federico II, Italy Serge Salat, UMCS Institute, France Mattheos Santamouris, NK University of Athens, Greece Ali Soltani, Shiraz University, Iran

#### Associate Editors

Rosaria Battarra, CNR, Italy Matteo Caglioni, Université Cote D'azur, France Alessia Calafiore, University of Edinburgh, UK Gerardo Carpentieri, University of Naples Federico II, Italy Luigi dell'Olio, University of Cantabria, Spain Isidoro Fasolino, University of Salerno, Italy Stefano Franco, Politecnico di Bari, Italy Federica Gaglione, University of Sannio, Italy Carmen Guida, University of Naples Federico II, Italy Thomas Hartmann, Utrecht University, Netherlands Markus Hesse, University of Luxemburg, Luxemburg Zhanat Idrisheva, D. Serikbayev EKTU, Kazakhstan Zhadyra Konurbayeva, D. Serikbayev EKTU, Kazakhstan Seda Kundak, Technical University of Istanbul, Turkey Rosa Anna La Rocca, University of Naples Federico II, Italy Houshmand Ebrahimpour Masoumi, TU of Berlin, Germany Giuseppe Mazzeo, Pegaso Telematic University, Italy Nicola Morelli, Aalborg University, Denmark Enrica Papa, University of Westminster, United Kingdom Yolanda P. Boquete, University of Santiago de Compostela, Spain Dorina Pojani, University of Queensland, Australia Nailya Saifulina, University of Santiago de Compostela, Spain Athena Yiannakou, Aristotle University of Thessaloniki, Greece John Zacharias, Peking University, China Cecilia Zecca, Royal College of Art, UK Floriana Zucaro, University of Naples Federico II, Italy

#### EDITORIAL STAFF

Laura Ascione, Ph.D. student at University of Naples Federico II, Italy Annunziata D'Amico, Ph.D. student at University of Naples Federico II, Italy Valerio Martinelli, Ph.D. student at University of Naples Federico II, Italy Stella Pennino, Ph.D. student at University of Naples Federico II, Italy Tonia Stiuso, Research fellowship at University of Naples Federico II, Italy

# TeMA Journal of Land Use, Mobility and Environment

### NEW CHALLENGES FOR XXI CENTURY CITIES:

Multilevel scientific approach to impacts of global warming on urban areas, energy transition, optimisation of land use and emergency scenario

1 (2025)

### Contents

EDITORIAL PREFACE 3 Rocco Papa

#### FOCUS

- Situating walkability examining walkability elements of recurring routes 7 Jani Tartia
- Definition of spatio-temporal levels of accessibility. Isochronous analysis of regional 23 transport networks Annunziata Palermo, Gaetano Tucci, Lucia Chieffallo
- The impact of transportation planning on agricultural areas and plant health: a case 39 study of Antalya/Konyaaltı West Ring Road Engin Kepenek, Engin Kepenek, Şerife Betül Çetinkaya

LUME (Land Use, Mobility and Environment)

- Campi Flegrei and the Metropolitan Area of Naples. 55 Emergency planning in a high-risk territory Giuseppe Mazzeo
- Revitalising abandoned historical districts. 79 Application of an incremental and adaptive approach to regeneration Diksha Dody, Daniele Ronsivalle, Maurizio Carta

- 95 Mobilising equity. Emerging evidence for integrating vulnerable communities Irina di Ruocco
- **113** Multilevel governance approach to adaptation. The construction of the Italian mid-Adriatic green infrastructure Timothy Daniel Brownlee, Rosalba D'Onofrio, Chiara Camaioni

#### **REVIEW NOTES**

- **131** Urban energy transition between regulatory evolution and scientific production: a bibliometric analysis Valerio Martinelli
- **143** Digitalization in urban planning: a framework to realize smart cities Annunziata D'Amico
- **151** Competitive climate adaptation. Italian start-ups leading the way to adaptation to climate change in cities Stella Pennino
- **161** Exploring open and green space characteristics for climate change adaptation: a focus on the urban heat island Tonia Stiuso
- **169** Global warming reports: a critical overview of IGOs publications Laura Ascione

# TeMA

Journal of Land Use, Mobility and Environment

TeMA 1 (2025) 95-112 print ISSN 1970-9889, e-ISSN 1970-9870 DOI: 10.6093/1970-9870/10872 Received 11<sup>th</sup> April 2024, Accepted 3<sup>rd</sup> February 2025, Available online 30<sup>th</sup> April 2025

Licensed under the Creative Commons Attribution – Non Commercial License 4.0 https://serena.sharepress.it/index.php/tema

## Mobilising equity. Emerging evidence for integrating vulnerable communities

#### Irina Di Ruocco

Department of Economics, University of Insubria, Varese, Italy e-mail: diruocco@uninsubria.it ORCID: https://orcid.org/0000-0003-0829-0754

#### Abstract

The transition towards sustainable cities is progressing, with numerous smart city technologies and innovative concepts being implemented. However, these advancements have predominantly focused on urban areas, often overlooking the critical need for integrating transportation systems with surrounding rural regions. Equitable access for vulnerable populations in peripheral areas remains underexplored, with most research focusing on the elderly while neglecting other vulnerable groups such as ethnic and gender minorities, women, young people, and individuals with serious health conditions. Empirical evidence shows that many rely on private transportation or are confined to small, localized areas. This article, through a literature review, analyses emerging research evidence on transportation for vulnerable groups, emphasizing the need to integrate technological solutions for individuals with reduced mobility (e.g. such as the visually impaired) into digital mobility platforms and land-use planning. The study highlights the lack of focus on rural regions and diverse vulnerable groups, stressing the importance of better integrating technology and land-use planning to improve transportation accessibility.

#### **Keywords**

Vulnerable groups; Accessibility; Mobility inclusive; Sustainable mobility; Policies.

#### How to cite item in APA format

Di Ruocco, I. (2025). Mobilising equity. Emerging evidence for integrating vulnerable communities. *TeMA - Journal of Land Use, Mobility and Environment, 18*(1), 95-112. http://dx.doi.org/10.6093/1970-9870/10872

#### 1. Introduction

The connection between transportation systems, land use, and mobility is fundamental for fostering social inclusion and well-being (Preston & Rajé, 2007). Efficient transportation access is essential for participation in key daily activities, such as healthcare, employment, and recreation (Gatrell, 2013; Papagiannidis et al., 2017). However, approximately 16% of the global population, including individuals with physical, sensory, or cognitive impairments, economically disadvantaged groups, and ethnic minorities, face restricted mobility options, limiting their ability to access essential services (WHO, 2021). This issue is expected to intensify with demographic changes, including aging populations and rising non-communicable diseases (He et al., 2020). Accessibility challenges persist, particularly in low-income and underserved areas (Martens, 2019), where infrastructure investments are minimal, leading to high travel costs and extended commute times (El-Geneidy et al., 2016). Despite technological advancements (Oguz & Tanyas, 2024), marginalized groups continue to face systemic barriers, including financial limitations, physical inaccessibility (Soltani et al., 2012), and discriminatory practices. Technologies such as Intelligent Control Technology "ICT" and Augmented Reality "AR" offer promising tools, but their adoption often fails to address the specific needs of diverse vulnerable populations, especially in rural contexts (Nixon & Schwanen, 2018; Ma et al., 2018). Moreover, the COVID-19 pandemic further exposed the vulnerability of transportation systems, particularly for the elderly (LaPlante, 2014), immunocompromised, and those experiencing intersecting forms of marginalization (Beck & Hensher, 2020). This highlights the urgent need for resilient and inclusive infrastructure that caters to diverse user needs.

Understanding mobility vulnerabilities requires an intersectional approach to examine how social identities like gender, race, disability, and economic status intersect, creating compounded barriers to transportation access (Crenshaw, 1991; Sheller, 2018). For example, low-income women may face financial, cultural, and safety challenges, while ethnic minorities or individuals with disabilities may encounter discrimination limiting their mobility. Addressing these inequalities demands targeted strategies that consider these overlapping factors. (D'Agostino et al., 2024).

However, empirical studies lack comprehensive evaluations of technology adoption rates among economically disadvantaged or elderly users. Yet, evidence on how new design changes have tangibly improved daily mobility outcomes for vulnerable populations remains limited. Globally, policy innovations and best practices (BPs) offer valuable lessons for addressing mobility challenges, yet their impacts remain underexplored. The selected examples stand out for their integration of inclusivity and technological innovation, demonstrating measurable outcomes while addressing physical and digital barriers to mobility. In Finland, the Whim app integrates diverse transport modes into a seamless, multi-modal platform with flexible fare structures, reducing travel barriers and enhancing convenience. However, scalability and affordability remain challenges. Barcelona's accessible metro system exemplifies universal design principles, improving accessibility for mobility-impaired users, though long-term maintenance remains critical. In Medellín, the Metrocable system connects low-income hillside communities with central hubs, reducing travel times and improving access to education and healthcare. However, questions about subsidy sustainability persist.

São Paulo's community-led BRT system emphasizes participatory planning, fostering community ownership and localized solutions, though consistent institutional commitment is essential. In London, the contactless payment system enhances efficiency and reduces boarding times but lacks comprehensive analysis of its impact on low-income users (Moore et al., 2013). Similarly, Los Angeles' electric bus initiative offers environmental benefits but requires deeper evaluation of equity outcomes.

These cases reveal key lessons: inclusivity must be embedded from the outset, technology serves as an enabler but must remain accessible, and long-term financial sustainability is essential. Additionally, community participation strengthens adoption and trust. While these BPs are context-specific, they offer adaptable frameworks for designing equitable, efficient, and sustainable mobility systems. Future research should focus on evaluating these dimensions to ensure broader applicability and long-term impact.

While these examples highlight promising interventions, they are often presented in fragmented narratives, lacking systematic evaluation frameworks to assess their long-term impacts on accessibility, social inclusion, and resilience. This paper addresses this analytical gap through a comprehensive literature review, incorporating academic and grey literature, including policy documents and case studies that explicitly focus on these dimensions but remain underexplored empirically.

The analysis focuses on three key dimensions (Litman, 2022; Lucas, 2011):

- Equity-focused policies addressing disparities related to age, health, gender, and economic status to ensure fair mobility access;
- b) Community engagement frameworks that align mobility solutions with the lived experiences of vulnerable groups, fostering trust and sustainable adoption;
- c) Technology adoption strategies designed to bridge urban-rural divides and promote inclusive digital accessibility.

This study contributes to a deeper understanding of how transportation systems can reduce mobility inequalities and enhance social inclusion, advocating for further empirical research to validate these insights across diverse geographical and socio-economic contexts.

The central research question (RQ) guiding this study is: "How can technological and digital mobility solutions for users with reduced mobility be integrated into transportation systems to enhance accessibility for vulnerable populations, particularly in underserved urban and rural areas?".

The study structure is as follows. After this introduction, Section 2 describes the empirical background on current trends. Section 3 outlines the methodology. Section 4 discusses the results and discussion. Section 5 concludes with recommendations for policymakers, urban planners, and stakeholders to create equitable, inclusive, and sustainable mobility systems.

#### 2. Background

The concept of social vulnerability includes not only "vulnerable populations" but also fragile individuals, such as the elderly with reduced mobility, the visually impaired, and the immunocompromised. Addressing their needs relies on technologies like Mobility-as-a-Service (MaaS), Artificial Intelligence (AI), and machine learning, which offer tailored navigation tools. However, these technologies remain only partially integrated into transportation systems, limiting their accessibility and effectiveness.

Despite advances in digital route assistance, elderly and visually impaired users still face significant barriers, especially in areas with poor technological infrastructure. Public transport systems often fail to accommodate immunocompromised individuals, pushing them to rely on private vehicles. In rural areas, and highlighted by O'Neill & O'Mahony (2005) for Ireland, the limited deployment of transport technologies further restricts mobility options and access to essential services.

While mobility planning increasingly prioritizes accessibility and equity, integrating trends such as ecological mobility and active transportation remains challenging for fragile groups. The adoption of ICT, MaaS, and Big Data technologies is still limited, preventing scalable and inclusive solutions.

The condition of mobility poverty reflects a lack of reliable, affordable, and sustainable transportation options, constrained by spatial, economic, technological, and cultural barriers (Kuttler & Moraglio, 2023). Overcoming these barriers requires integrated, equity-focused strategies to ensure accessible mobility for all.

From a spatial and geographic perspective, mobility barriers are distributed unevenly, with rural and periurban areas suffering from inadequate infrastructure and unreliable transit networks (O'Neill & O'Mahony, 2005). These deficits often force residents to rely on private vehicles, increasing financial burdens and environmental degradation. In contrast, urban centers face different but equally challenging issues, including overcrowded networks and limited last-mile connectivity.

The economic dimension highlights the burden of transportation costs on low-income populations, who often cannot afford private vehicles or costly urban transit fares. Maintenance expenses, combined with the financial pressures of transit access, exacerbate existing inequalities (Lucas, 2011). These economic barriers frequently overlap with spatial disparities, intensifying social exclusion (Ignaccolo et al., 2016).

In the technological sphere, innovations like Mobility-as-a-Service (MaaS) and AI-driven mobility tools offer solutions for optimizing transit routes, improving flexibility, and enabling adaptive services. Multiple transport services into a single platform, simplifying travel for users (Sen et al., 2022). However, challenges such as digital literacy gaps, limited infrastructure readiness, and financial barriers prevent widespread adoption, especially in rural and underserved areas (Park et al., 2022).

The environmental dimension focuses on the sustainability challenges associated with mobility poverty. In regions where public transport is sparse or unreliable, dependency on private vehicles leads to increased carbon emissions and worsened air quality. At the same time, sustainable infrastructure—such as cycling lanes and pedestrian pathways—remains underdeveloped in vulnerable regions, further limiting accessible options (Titheridge et al., 2014).

The social and cultural dimension reflects barriers rooted in discrimination, safety concerns, and cultural exclusion, particularly for women, ethnic minorities, and migrants. Poorly lit transit stops, limited language assistance, and systemic biases in planning disproportionately impact these groups, reinforcing their exclusion from mobility networks. According to D'Agostino et al. (2024) women in advanced countries tend to travel shorter distances, make more daily trips, use cars less frequently, and are more willing to adopt sustainable travel behaviours than men.

To address these interconnected barriers, land-use planning and transportation systems must be integrated effectively. Mixed-use development reduces travel distances to essential services, while Transit-Oriented Development (TOD) enhances connectivity and encourages reliance on sustainable transit options. Cerdá et al. (2012) describing Medellín's Metrocable system, demonstrate the potential of TOD policies to improve spatial connectivity and inclusion. However, without complementary housing and affordability policies, TOD strategies risk triggering gentrification and displacement, ultimately excluding the very populations they aim to serve (Vecchio & Martens, 2021).

Technological advancements like MaaS platforms and AI tools offer additional opportunities to bridge mobility gaps. For individuals with motor, sensory, or cognitive impairments, MaaS platforms provide real-time updates on accessible routes, while AI-driven systems optimize service delivery and route planning to accommodate demand fluctuations (Park et al., 2022). However, these technologies remain underutilized in rural and marginalized areas, where affordability and digital literacy gaps persist. Evidence in European cities (e.g., Toronto and Barcelona) shows how policy initiatives can reduce financial and infrastructure barriers. Toronto's fare subsidy has notably improved transit affordability for low-income users (Barri et al., 2021), while Barcelona's station redesigns increased transit usability for individuals with mobility impairments by 27% (Comim, 2008). Nevertheless, both cases emphasize the importance of sustained investment and community engagement to ensure long-term success. Examples like London's contactless payment system and Los Angeles' electric bus initiative highlight how technology and environmental goals can intersect with social equity, albeit with limitations in reaching marginalized communities (Bezbradica & Ruskin, 2019; Allen & Farber, 2020).

Addressing mobility poverty requires a multi-faceted strategy that combines targeted technological integration, policy alignment, and sustained infrastructure investment. The next section will build upon these insights,

presenting evidence-based findings from the literature to highlight effective strategies for overcoming mobility barriers and fostering inclusive transportation systems.

Building on the earlier discussion, social vulnerability and mobility poverty are deeply interconnected, as limited transportation access amplifies exclusion from economic, social, and cultural activities. Vulnerable groups, including older adults, individuals with physical, sensory, or cognitive impairments, economically disadvantaged populations, women (D'Agostino et al., 2024), youth, and ethnic minorities, face compounded barriers due to inadequate infrastructure, affordability constraints, and systemic exclusion (Battarra et al., 2018; Park & Chowdhury, 2018).

Globally, well-planned transportation systems have shown potential to improve access to employment, healthcare, and education, fostering social inclusion and economic growth (Cerdá et al., 2012; Vecchio & Martens, 2021). Investments in accessible metro systems and fare equity programs demonstrate positive outcomes, but their scalability remains limited (Comim, 2008; Barri et al., 2021).

Technological innovations, including Mobility-as-a-Service (MaaS) and digital payment systems, offer promising solutions. However, while affluent users benefit from seamless integration, low-income populations often face financial barriers, limiting their accessibility and equity impact (Arai et al., 2022).

Addressing these challenges requires integrated strategies that bridge spatial, economic, and technological divides to ensure inclusive, accessible, and resilient mobility systems. Further empirical research is essential to validate these approaches and assess their effectiveness across diverse contexts.

#### 3. Methodology

#### 3.1 Identification of target groups

In line with the literature review methodology described, this section identifies the key vulnerable groups facing mobility barriers and transport poverty, drawing from established literature (Hine & Mitchell, 2016; Lucas et al., 2016; Van Wee & Geurs, 2011). These studies emphasize how systemic inequalities, infrastructural limitations, and socio-economic disparities create persistent mobility challenges for marginalized populations.

The specific user groups analysed include:

1. Individuals with Disabilities: People with mobility, sensory, or cognitive impairments who encounter physical and infrastructural barriers within transportation systems (Currie et al., 2010);

2. Economically Disadvantaged Populations: Communities with limited financial resources, often disproportionately affected by high transport costs and underfunded infrastructure (Lucas, 2012);

3. Elderly Populations: Older adults whose physical and cognitive limitations, coupled with transport design barriers, restrict their independent mobility (Shergold & Parkhurst, 2012);

4. Women and Youth: Groups facing social and cultural barriers, including safety concerns, caregiving responsibilities, and financial dependence (Uteng & Cresswell, 2008);

5. Ethnic Minorities and Marginalized Communities: Populations experiencing systemic discrimination and cultural exclusion in accessing transportation services (Soja, 2013; Oviedo & Sabogal, 2020);

6. Residents in Rural or Remote Areas: Individuals in low-density regions where public transport services are often unreliable, sparse, or economically unfeasible (Nutley, 2003).

As shown in Tab.1, these user groups were selected based on their documented vulnerabilities in global policy discussions on mobility justice, their prominence in academic research, and their alignment with sustainable transport strategies (Jones et al., 2018; Lucas, 2012). These references highlight the interconnected nature of social equity, mobility access, and transport infrastructure investment, providing a robust foundation for

this analysis. These groups represent the focus of the reviewed studies and provide a foundation for analyzing mobility-related social exclusion and accessibility challenges. Tab.1 summarizes their barriers, spatial contexts, and rationales:

- 1. Low-income and Unemployed Individuals:
- Barrier. High transportation costs, limited public transport access;
- Spatial Context. Urban reliance on public transit; rural dependence on car ownership;
- Rationale. Transport costs exacerbate poverty and reliance on inefficient systems (Lucas et al., 2016).
  In 2022, 21.7% of the EU population faced poverty risk (Eurostat, 2022).
- 2. Elderly Persons, Women, and Gender-Diverse Groups:
- Barrier. Physical infrastructure issues, safety concerns, limited services;
- Spatial Context. Urban and rural areas, with acute challenges in low-density zones;
- Rationale. Women face safety concerns, while elderly individuals encounter physical barriers. In 2021, 22.7% of women in the EU faced poverty risk (Eurostat, 2022).
- 3. Young People (16-24 years):
- Barrier. Financial dependence, restricted access to transport options;
- Spatial Context. Urban and suburban areas with infrequent transit services;
- Rationale. Limited financial independence hinders youth mobility (Eurostat, 2022).
- 4. Migrants and Ethnic Minorities:
- Barrier. Language barriers, financial constraints, cultural discrimination;
- Spatial Context. Predominantly urban areas with uneven transport access;
- Rationale. Reduced access due to financial and language challenges. In 2021, 5.3% of the EU population were non-EU citizens (Eurostat, 2022).
- 5. People with Reduced Mobility:
- Barrier. Inadequate infrastructure, lack of adapted services, technological exclusion;
- Spatial Context. Urban architectural barriers; poor rural services;
- Rationale. Specially adapted services are essential for equitable mobility. In 2021, 29.7% of disabled individuals in the EU faced poverty risk (Eurostat, 2022).
- 6. Residents of Rural, Disadvantaged, and Marginal Urban Areas:
- Barrier. Sparse networks, low transit frequency, long travel times;
- Spatial Context. Rural and peripheral areas with limited services;
- Rationale. Poor infrastructure limits access to essential services. In 2018, 29.1% of EU residents lived in rural areas (Eurostat, 2022).
- 7. Immunocompromised Individuals:
- Barrier. Safety and hygiene concerns in shared transport systems;
- Spatial Context. Urban and rural environments, with higher risks in crowded settings;
- Rationale. Safe travel conditions are essential but often unmet (Pezzagno & Richiedei, 2022).

These groups highlight the multifaceted barriers to mobility, emphasizing the need for targeted strategies to address their unique challenges across urban, suburban, and rural contexts.

#### 4. Results and discussion

This section describes the results of the methodology, highlighting the role of policy-driven mobility solutions in overcoming barriers faced by older adults, individuals with impairments, economically disadvantaged populations, women, youth, and ethnic minorities.

Keyword Sector	Target Groups	Area of Interest	Quantitative Methods	Qualitative Methods	Focus of Vulnerability	Data Collection Method	Example References	No. of Papers Reviewed
Land-Use Integration	Economically disadvantaged groups	Transport policies, geography	$\checkmark$	$\checkmark$	Socio-economic exclusion	Travel survey, GIS	El-Geneidy et al., 2016; Welch & Mishra, 2013	12
Gender and Youth Mobility	Women, Youth	Behavioral science	$\checkmark$	$\checkmark$	Gender-specific barriers, safety	GPS tracking, Interviews	Vecchio & Martens, 2021; Park et al., 2022	10
Assistive Transport Tools	Individuals with impairments	Health, accessibility	$\checkmark$	$\checkmark$	Mobility, cognitive, sensory barriers	Questionnaire, Interviews	Li & Loo, 2017; Park & Chowdhury, 2018	8
Marginalized Communities	Ethnic minorities	Co-design, urban planning	$\checkmark$	$\checkmark$	Systemic and cultural exclusion	Travel survey, Focus groups	Nixon & Schwanen, 2018; Hananel & Berechman, 2016	9
Spatial Accessibility	Rural and remote communities	Regional geography	$\checkmark$	$\checkmark$	Infrastructure and distance barriers	Travel survey, GIS	Combs et al., 2016; Papa et al., 2018	7
Inclusive Policy Frameworks	Elderly	Social inclusion policies	$\checkmark$	$\checkmark$	Age-related accessibility barriers	Community workshops	Battarra et al., 2018; Shareck et al., 2014	6
Regulatory Policies	Cross-cutting (multiple groups)	Regulation, urban policy	$\checkmark$	$\checkmark$	Institutional and policy gaps	Document analysis	Di Ciommo & Shiftan, 2017; Allen & Farber, 2020	13

 $\checkmark$ : Indicates that studies within the category utilized the specified methodological approach.

Tab.1 Overview of Reviewed Studies by Keyword Sector, Target Groups, and Methodologies (Elaboration of author)

The key findings of this research align with the five thematic dimensions (4.1–4.5) and offer an integrated framework for addressing mobility challenges:

- 1. Equitable transportation systems are essential for improving access to social, economic, and cultural opportunities. This aligns with Section 4.1: Policy-Oriented Mobility Targeting Vulnerable Groups, which emphasizes the role of targeted policy interventions, and Section 4.2: Accessibility and Proximity in Smart Cities, which focuses on spatial planning and urban design to minimize mobility barriers;
- Active community participation ensures mobility solutions are aligned with the lived experiences and needs of end users. Section 4.5: Technology and Community Engagement explores participatory approaches to policymaking, while Section 4.4: Spatial and Infrastructure Resilience underscores how community input strengthens adaptability and responsiveness in infrastructure development;
- 3. Innovations such as Mobility-as-a-Service (MaaS) and ICT platforms offer scalable solutions to bridge mobility gaps. Section 4.3: Mobility Technologies for Vulnerable Groups examines these technologies' role in improving transport efficiency and inclusivity, while Section 4.4: Spatial and Infrastructure Resilience highlights the importance of context-specific implementation to address rural-urban disparities.

Together, these dimensions—policy frameworks (4.1), spatial planning (4.2), technological integration (4.3), infrastructure resilience (4.4), and community engagement (4.5) - form an interconnected strategy for combating mobility poverty. Each dimension contributes uniquely: policy provides direction, spatial planning ensures physical accessibility, technology drives innovation, resilient infrastructure sustains long-term access, and community engagement guarantees inclusivity and relevance. The following section builds on these insights, offering specific recommendations and actionable pathways to implement inclusive mobility solutions effectively. Lastly, the discussion proposes also an evaluation of different challenges across the types of regions, and the future development for research (4.7).

#### 4.1 Policy oriented mobility targeting for vulnerable groups

Policy-oriented mobility focuses on tailored transport policies designed to address the unique needs of vulnerable groups. For instance, Toronto's Fair Pass program subsidizes public transit for low-income users,

reducing financial barriers (Barri et al., 2021), while Barcelona's metro upgrades improved usability for individuals with impairments by 27% (Comim, 2008).

However, policy effectiveness depends on clear metrics for accessibility and ongoing evaluation. Initiatives like Finland's Whim app demonstrate how MaaS platforms can offer integrated services, but questions remain about who drives these initiatives and how accessibility outcomes are measured (Sen et al., 2022).

Emerging trends in urban mobility, as outlined in the "POLIS Report" (Polis, 2021) and the EIT Report (EIT Urban Mobility, 2021), emphasize integrating resilience into sustainable urban mobility planning. These reports explore strategies for post-COVID mobility systems, highlighting best practices from European cities (Budd & Ison, 2020; Pan & He, 2022; Beck & Hensher, 2020). A key focus is on inclusiveness and integration, ensuring vulnerable groups are considered in urban and rural transport planning. Public transport remains vital for people with special needs, providing access to jobs, education, and social opportunities in disadvantaged areas. The MaaS sector is exploring accessibility measures, such as the Urban Mobility Index and Deloitte's City Mobility Index (Dixon et al., 2019), but these have yet to fully address the needs of vulnerable users. EU guidelines (2014) emphasize accessibility through user-centered design, including digital devices, audible/visual announcements, and affordability. The co-creation approach encourages civic participation, allowing communities to co-identify needs, co-develop solutions, and co-evaluate outcomes (EU, 2019). Community policies focus on reducing digital exclusion by improving access to technology, enhancing digital literacy, and ensuring adequate infrastructure. The European Accessibility Act (Directive (EU) 2019/882 (European Union - Regulation 2019) and the Regulation on Technical Specifications for Interoperability (EU, 2014) further support accessibility in transportation for people with disabilities.

#### 4.2 Accessibility and proximity of smart cities

In smart city planning, accessibility serves as a core metric for evaluating the connection between people, territories, and transportation systems, while proximity, mobility, and connectivity represent the tools and processes enabling access. Accessibility, often considered an end goal, reflects the ability of individuals to reach essential services and opportunities efficiently and equitably, whereas mobility represents the movement itself, and proximity indicates the spatial arrangement of services relative to users.

Successful accessibility planning requires a shift from mobility-centered approaches to access-oriented frameworks, where the emphasis is placed on the capacity of transportation systems to meet social equity objectives. For example, Bocarejo & Oviedo (2012) developed a tool to analyze transport accessibility and social inequities, demonstrating how infrastructure and costs often limit access despite geographical proximity. Similarly, Sochor (2015) highlighted access deprivation through semi-structured interviews with disadvantaged populations, emphasizing the need to address both physical and economic barriers to access.

At the urban scale, Transit-Oriented Development (TOD) remains a widely adopted strategy to improve proximity and spatial accessibility by clustering housing, employment, and services around transit hubs. Medellín's Metrocable system stands as a notable example, successfully connecting marginalized neighborhoods with central urban areas (Xia et al., 2016) and significantly reducing travel times (Cerdá et al., 2012). However, TOD strategies can unintentionally lead to gentrification and displacement if not paired with complementary housing affordability policies (Vecchio & Martens, 2021). This highlights the necessity of cross-sectoral integration between transport, housing, and urban planning policies to ensure that improved accessibility translates into long-term inclusivity.

A critical issue in accessibility studies lies in the gap between theoretical proximity and real-world access. Pucci et al. (2019) note that even when facilities are geographically nearby, infrastructure quality, service availability, and costs often create significant barriers for vulnerable populations. To address these gaps, Paiva et al. (2022) argue for a social justice-oriented approach to accessibility, prioritizing equitable access across diverse socio-economic and demographic groups.

Measuring accessibility remains a persistent challenge, with methodologies varying significantly across studies. For instance, Ryan & Pereira (2021) and Sabella & Bezyak (2019) highlight the complexity of quantifying access deprivation, particularly for individuals with disabilities. Wong (2018) explores how social factors influence mobility among visually impaired individuals, while Boisjoly and El-Geneidy (2017) reviewed metropolitan transport plans, observing a trend towards integrating accessibility objectives. However, the practical application of accessibility-based indicators remains limited in many cities.

In terms of operational strategies, Cheng & Chen (2015) propose a multidimensional measure combining accessibility, mobility, and connectivity to evaluate public transport systems. Similarly, Guzman (Guzman et al., 2017; Guzman & Oviedo, 2018) examined 'pro-poor' public transport subsidies in Bogotá, showing how well-targeted financial measures can reduce access inequalities. Hidayati et al. (2019) report similar findings in Indonesia, underscoring the importance of context-specific policies in addressing affordability gaps.

In cities facing ageing population dynamics, accessibility becomes increasingly vital to support active ageing and continued participation in social and economic activities. Urban mobility systems must be designed to accommodate the specific needs of older adults, including considerations for safety, affordability, and physical accessibility. Initiatives integrating Smart City solutions, ICT technologies, and mobility services show promise in enhancing accessibility while addressing systemic vulnerabilities (Guida et al., 2020; Battarra et al., 2018; Pezzagno & Richiedei, 2022).

#### 4.3 Mobility technologies for vulnerable groups

Technological innovations, including Mobility-as-a-Service (MaaS) platforms and AI-driven systems, hold significant potential to enhance accessibility, adaptability, and efficiency in transportation networks for vulnerable groups. MaaS platforms, such as Finland's Whim app, integrate multiple transportation modes— public transit, bike-sharing, and ride-hailing—into a single digital interface, simplifying travel planning and fare payments while offering flexible mobility options (Sen et al., 2022). Similarly, AI-powered tools enable demand-responsive transit services, optimizing routes and schedules based on real-time travel patterns and user demand, improving service availability in underserved areas (Park et al., 2022). However, despite these advancements, technological adoption remains uneven, particularly in rural and marginalized areas, where affordability, infrastructure gaps, and digital literacy barriers persist.

The limited success of MaaS in rural regions often stems from low population densities, making it economically unviable for private operators to deploy large-scale services. In contrast, urban environments with higher user volumes have seen more successful MaaS adoption, as demonstrated by Vienna's Smile app, which integrates multimodal travel and offers tailored options for elderly and disabled users. Similarly, in China (Niehaus et al., 2016) and in Singapore, AI technologies have been embedded in public transit systems to predict congestion, optimize routes, and ensure accessibility for users with limited mobility. These examples highlight the need for context-specific implementation strategies that address geographic, infrastructural, and socio-economic barriers.

For elderly individuals, technology offers both opportunities and challenges. While ICT tools can improve independence and enable real-time route planning, older adults often face digital literacy barriers and struggle with interfaces designed without their needs in mind. Moreover, outdated infrastructure—such as poorly designed bus stops, uneven sidewalks, and inadequate signage—further undermines the potential benefits of mobility technologies (Pretty et al., 2002). Research by Riazi et al. (2016) underscores these challenges, identifying outdoor navigation difficulties for visually impaired individuals, including unreadable signs, disorientation, and safety hazards.

For individuals with disabilities, paratransit services remain a critical, albeit often limited, option. Studies by Egger et al. (2022) in Switzerland highlight the barriers and facilitators affecting paratransit adoption, including poor scheduling reliability, limited coverage, and inadequate integration with broader transit networks. Addressing these issues requires infrastructure improvements alongside technological advancements, such as

real-time scheduling apps and accessible payment systems, which have shown success in cities like Berlin and Zurich.

The digital divide is another significant barrier, particularly in rural areas, where access to affordable mobile devices and stable internet connections is limited. Programs like India's Common Service Centers (CSCs) demonstrate how localized ICT hubs can address digital literacy gaps, providing community-based access to digital transportation services.

Integration between technological solutions and urban planning remains a critical gap in both research and practice. Technologies like MaaS and AI-based mobility platforms are often implemented as isolated solutions rather than being embedded into broader transportation and land-use strategies. For example, while MaaS platforms in cities like Helsinki and Vienna offer flexible urban mobility options, their success relies on integrated land-use policies that ensure equitable coverage across diverse neighborhoods.

To maximize the benefits of mobility technologies for vulnerable populations, a holistic approach is essential. This includes affordable and context-specific digital solutions, inclusive design principles, infrastructure readiness, and community involvement in the design and deployment phases. Cities must not only invest in cutting-edge technologies but also ensure these systems are user-friendly, widely accessible, and tailored to the unique challenges faced by vulnerable populations.

#### 4.4 Spatial and infrastructure resilience

Resilient infrastructure ensures long-term, adaptive transit networks that respond to economic, environmental, and social challenges. Initiatives like Los Angeles' electric bus fleet showcase environmental sustainability but face challenges in reaching peripheral neighborhoods (Allen & Farber, 2020).

Similarly, São Paulo's community-led BRT systems highlight how local participation can improve infrastructure relevance and adoption (Cordoba et al., 2014). Success depends on sustained investment and adaptive planning.

Recent focus on resilience in urban transportation emphasizes reducing impacts and maintaining mobility, particularly for vulnerable populations (Wan et al., 2018; Mattsson & Jenelius, 2015). Resilience is closely tied to vulnerability and has gained attention in transportation studies, especially regarding reduced mobility for the elderly and visually impaired. Muller et al. (2021) identified mobility indices for MaaS, using heat maps to evaluate accessibility and performance. Van Dülmen et al. (2022) examined transportation poverty through GPS tracking in rural Czech and German regions, finding that social disadvantage often outweighs regional differences in mobility. Compagnucci & Morettini (2020) proposed a space-equity approach for equal access to services in inner regions, emphasizing the need for integrated systems, as in rural areas (Vitale Brovarone, 2021). Universal public transportation design often neglects vulnerable users, leading to unnecessary detours and collision risks, particularly for wheelchair users (Velho, 2019), the elderly, and women with strollers. Hranický et al. (2021) addressed rail transport for people with disabilities, focusing on specialized train cars. Arai et al. (2022) used network analysis to evaluate wheelchair accessibility at train stations, highlighting proximity, reachable times, and collision risks on designated paths.

#### 4.5 Technology and community engagement

Community engagement bridges policy intent with user needs, ensuring mobility solutions reflect lived experiences. Participatory initiatives, as seen in Barcelona's inclusive mobility plans and Medellín's communitydriven transit projects, demonstrate how local involvement fosters trust and alignment (Comim, 2008; Cerdá, et al., 2012).

However, engagement must go beyond consultation, requiring active citizen participation in planning, monitoring, and evaluation processes.

Research by Lee et al. (2022) in South-Corea identified a strong link between life satisfaction and perceived mobility environments among wheelchair users, highlighting the importance of social initiatives to boost mobility and social interaction (Dennis et al., 2016). European policies, such as those examined by Chiscano & Darcy (2022) in Barcelona, advocate for inclusive public transport co-designed with input from people with disabilities. Technologies can also address more nuanced factors like perceptions of accessibility, safety, and communication, all of which influence mobility decisions (Sochor, 2014). Though digitization has been slow among the elderly, the increasing use of ICT and Big Data supports a shift towards green, multimodal mobility solutions (Sochor, 2015). Navigation assistance for visually impaired individuals, such as the DeepNAVI system, shows potential for promoting independent living (Kuriakose et al., 2020). However, digital solutions sometimes introduce biases that restrict accessibility for vulnerable populations, including people with disabilities (Kwakye, 2022). Common barriers include administrative restrictions, negative driver attitudes, and difficulties in securing wheelchairs in vehicles (Hezam et al., 2022; Bezyak et al., 2019). Mora et al. (2017) emphasize the importance of involving vulnerable citizens as active participants in shaping more accessible urban environments (Ferrari et al., 2014). Cities can adopt a framework that combines urban planning, mobility strategies, and governance approaches (Martinelli, 2024) to enhance urban accessibility for the elderly (D'Amico, 2024), promoting an active, inclusive, and age-friendly urban environment (Gargiulo et al., 2018).

#### 4.6 Investigation of ICT solutions across regions

This section explores regional differences in the adoption and effectiveness of ICT-based mobility solutions for vulnerable groups, highlighting distinct challenges and opportunities in urban and rural contexts as identified in the literature. In urban areas, ICT tools such as Mobility-as-a-Service (MaaS) platforms, AI-powered route optimization, and digital payment systems have shown significant potential to improve accessibility and reduce mobility barriers. For example, Finland's Whim app integrates multiple transport modes into a single platform, offering flexibility and convenience (Sen et al., 2022). Similarly, Singapore employs AI-driven systems to predict congestion and optimize routes, enhancing accessibility for vulnerable users (Arai et al., 2022; Zhou et al., 2012). Research by Battarra et al. (2018) emphasizes how smart mobility technologies, when paired with inclusive urban planning, can support accessibility for the elderly and people with disabilities. However, Wong (2018) and Bezyak et al. (2019) highlight persistent challenges, including digital literacy gaps and interface design shortcomings, which often exclude older adults and visually impaired individuals. In rural regions, ICT adoption faces systemic barriers, including low population density, poor digital infrastructure, and limited financial sustainability (Combs et al., 2016; Milbourne & Kitchen, 2014). Studies by Egger et al. (2022) reveal that paratransit services, often supported by ICT scheduling tools, struggle with limited coverage and operational inefficiencies. Additionally, the digital divide remains a critical obstacle, as access to affordable mobile devices and reliable internet connections is often restricted (AbuJarour, 2022; Pezzagno & Richiedei, 2022). Cross-regional analysis reveals that affordability and user-centered design are central to the effectiveness of ICT solutions. Research by Chiscano & Darcy (2022) stresses the importance of co-creation approaches in developing accessible transport technologies, while Busco et al. (2023) argue that financial barriers often prevent low-income populations from fully benefiting from subscription-based MaaS platforms. Holistic strategies integrating ICT solutions with broader transport policies and community engagement frameworks are essential. Allen & Farber (2020) propose an accessibility-activity participation model to evaluate social inclusion outcomes, while Bantis & Haworth (2020) suggest adopting a capabilities-based approach to assess ICT interventions' impacts on equity. Additionally, Gargiulo et al. (2018) emphasize the need for smart city frameworks that prioritize social justice and ensure that technological advancements do not exacerbate existing inequalities. In conclusion, while ICT-based mobility solutions offer transformative potential, their success depends on affordable digital tools, localized infrastructure investments, inclusive design, and active community participation to address disparities across urban and rural contexts (Boisjoly & El-Geneidy, 2017; Hananel & Berechman, 2016).

#### 4.7 Further research directions

Equitable and accessible transportation systems must address the physical, economic, and social barriers faced by vulnerable groups, including the elderly, disabled, and economically disadvantaged (Stanley & Stanley, 2017; Stanley et al., 2011, 2012, 2019, 2022). Infrastructure improvements, such as ramps, accessible vehicles, and ICT tools for journey planning, are essential for enhancing mobility equity. Technologies like Mobility-as-a-Service (MaaS) must be user-friendly and tailored to accommodate diverse abilities (Battarra et al., 2018).

In rural areas, where mobility barriers are more pronounced, specific investments are needed to reduce isolation and improve access to essential services such as healthcare and education (Di Ruocco, 2022a,b, 2024; Combs et al., 2016). Integration between land use planning and transport systems can minimize travel distances and improve accessibility (Lucas, 2012; Kuttler & Moraglio, 2023). Transit-Oriented Development (TOD) principles, typically urban-focused, should be adapted for rural hubs to create better connectivity.

Technological solutions, including MaaS and ICT platforms, have potential but require adaptability and scalability to overcome rural infrastructure limitations (Arai et al., 2022; Zhou et al., 2012). Additionally, digital literacy programs and affordable digital tools must accompany these solutions to bridge existing divides (AbuJarour, 2022; Bezyak et al., 2019).

Community engagement is critical for developing transport solutions that reflect local needs. Inclusive participatory planning fosters trust and aligns infrastructure with social equity goals (Chiscano & Darcy, 2022; Papa et al., 2018). Coordinated approaches involving governments, private sectors, and NGOs can ensure sustained implementation and policy alignment (Allen & Farber, 2020; Bantis & Haworth, 2020).

On this premise, this study proposes future research directions:

- Comparative Regional Analysis: Assess how urban and rural settings respond differently to MaaS and ICT adoption, focusing on factors such as infrastructure readiness, financial barriers, and cultural acceptance;
- Spatial Integration Studies: Examine how land use policies and transportation networks can be better aligned to improve accessibility across diverse geographic contexts;
- Technological Inclusivity: Investigate how digital tools can be optimized for elderly and disabled users, addressing challenges in interface design and usability;
- Affordability and Financial Models: Explore sustainable funding mechanisms, including fare subsidies and public-private partnerships, to ensure equitable access to mobility services;
- Case Study Evaluation: Deep dive into best practices from successful transport initiatives, such as Toronto's Fair Pass program or Medellín's Metrocable system, to identify replicable strategies (Cordoba et al., 2014; Dharmowijoyo & Joewono, 2019).

Policy Effectiveness Metrics: Develop measurable indicators to evaluate the social, economic, and environmental impacts of transport interventions in underserved areas.

#### 5 Conclusions

The findings suggest that while technologies like Mobility-as-a-Service (MaaS) and ICT offer significant potential for enhancing mobility, their effectiveness is constrained without the support of well-integrated land use planning and inclusive policies (Shliselberg & Givoni, 2017). Addressing mobility poverty requires a comprehensive approach that includes technological innovations, sustainable land use strategies, and policies that promote inclusion and equity. This integrated approach will help reduce transportation disparities and improve access to essential services, particularly for vulnerable populations in both urban and rural areas. Expanding on these strategies can foster a more equitable transportation system that prioritizes social inclusion and sustainability (Pezzagno & Richiedei, 2022). However, limitations include an urban-centric focus in the literature, limited insights from rural areas, and a lack of longitudinal data to measure long-term impacts.

Despite these gaps, the study distinguishes disability-specific barriers from broader socio-economic challenges, ensuring a clear analysis of mobility inequities (Remillard et al., 2022). It highlights the need for affordable digital tools, context-specific policies, and community engagement to bridge gaps in accessibility.

Future research will refine these insights, focusing on successful case studies and developing measurable indicators for evaluating policy effectiveness. This study offers a foundation for inclusive, resilient, and equitable transportation systems, providing actionable insights for policymakers and stakeholders

#### References

AbuJarour, S. (2022). Integration through education: Using ICT in education to promote the social inclusion of refugees in Germany. *Journal of Information Systems Education*, *33*(1), 51-60. https://aisel.aisnet.org/jise/vol33/iss1/7

Allen, J. & Farber, S. (2020). Planning transport for social inclusion: An accessibility-activity participation approach. *Transportation Research Part D: Transport and Environment, 78*, 102212. https://doi.org/10.1016/j.trd.2019.102212

Arai, Y., Kusakabe, T., Niwa, Y. & Honma, K. (2022). Evaluation of wheelchair accessibility in train stations using a spatial network. *Asian Transport Studies, 8*, 100067. https://doi.org/10.1016/j.eastsj.2022.100067

Bantis, T. & Haworth, J. (2020). Assessing transport-related social exclusion using a capabilities approach to accessibility framework: A dynamic Bayesian network approach. *Journal of Transport Geography, 84*, 102673. https://doi.org/10.1016/j.jtrangeo.2020.102673

Barri, E. Y., Farber, S., Kramer, A., Jahanshahi, H., Allen, J. & Beyazit, E. (2021). Can transit investments in low-income neighbourhoods increase transit use? Exploring the nexus of income, car-ownership, and transit accessibility in Toronto. Transportation Research Part D: Transport and Environment, 95, 102849. https://doi.org/10.1016/j.trd.2021.102849

Battarra, R., Zucaro, F. & Tremiterra, M. (2018). Smart mobility and elderly people. Can ICT make the city more accessible for everybody?. *TeMA - Journal of Land Use, Mobility and Environment*, 23-42. https://doi.org/10.6092/1970-9870/5768

Beck, M. J. & Hensher, D. A. (2020). Insights into the impact of COVID-19 on household travel and activities in Australia– The early days under restrictions. *Transport Policy, 96*, 76-93. https://doi.org/10.1016/j.tranpol.2020.07.001

Bezbradica, M. & Ruskin, H. J. (2019). Understanding urban mobility and pedestrian movement. In V. Bobek (Ed.), *Smart Urban Development*, 149. IntechOpen eBooks. https://doi.org/10.5772/intechopen.86801

Bezyak, J. L., Sabella, S., Hammel, J., McDonald, K., Jones, R. A. & Barton, D. (2019). Community participation and public transportation barriers experienced by people with disabilities. *Disability and Rehabilitation, 42* (23), 3275-3283. https://psycnet.apa.org/doi/10.1080/09638288.2019.1590469

Bocarejo S., J. P. & Oviedo H., D. R. (2012). Transport accessibility and social inequities: A tool for identification of mobility needs and evaluation of transport investments. *Journal of Transport Geography, 24*, 142–154. https://doi.org/10.1016/j.jtrangeo.2011.12.004

Boisjoly, G. & El-Geneidy, A. M. (2017). How to get there? A critical assessment of accessibility objectives and indicators in metropolitan transportation plans. *Transport Policy, 55*, 38-50. http://dx.doi.org/10.1016/j.tranpol.2016.12.011

Budd, L. & Ison, S. (2020). Responsible transport: A post-COVID agenda for transport policy and practice. *Transportation Research Interdisciplinary Perspectives, 6*, 100151. http://dx.doi.org/10.1016/j.trip.2020.100151

Busco, C., González, F., Carmona, P., Cancino, A. & Alonso, B. (2023). Social exclusion and the public bus system in Santiago, Chile. *Cities, 132*, 104092. http://dx.doi.org/10.1016/j.cities.2022.104092

Cerdá, M., Morenoff, J. D., Hansen, B. B., Tessari Hicks, K. J., Duque, L. F., Restrepo, A. & Diez-Roux, A. V. (2012). Reducing violence by transforming neighborhoods: a natural experiment in Medellín, Colombia. *American journal of epidemiology*, *175*(10), 1045-1053. https://doi.org/10.1093/aje/kwr428

Cheng, Y. & Chen, S. (2015). Perceived accessibility, mobility, and connectivity of public transportation systems. *Transportation Research Part A: Policy and Practice*, *77*, 386-403. https://doi.org/10.1016/j.tra.2015.05.003

Chiscano, M. C. & Darcy, S. (2022). An accessible and inclusive public transportation management response to COVID-19 through a co-creation process with people with disability. The case of Metro Barcelona. *Research in Transportation Business & Management, 45*, 100880. https://doi.org/10.1016/j.rtbm.2022.100880

Combs, T. S., Shay, E., Salvesen, D., Kolosna, C. & Madeley, M. (2016). Understanding the multiple dimensions of transportation disadvantage: The case of rural North Carolina. *Case Studies on Transport Policy*, *4* (2), 68-77. http://dx.doi.org/10.1016/j.cstp.2016.02.004

Comim, F. (2008). Measuring capabilities. In F. Comim, M. Qizilbash & S. Alkire (Eds.), *The Capability Approach: Concepts, measures and applications*. Cambridge University Press.

Compagnucci, F. & Morettini, G. (2020). Improving resilience at the local level: The location of essential services within inner areas. Three case studies in the Italian Marche region. *Regional Science Policy & Practice, 12* (5), 767-793. https://doi.org/10.1111/rsp3.12343

Cordoba, D. Z., Stanley, J. & Stanley, J. R. (2014). Reducing social exclusion in highly disadvantaged districts in Medellín, Colombia, through the provision of a cable-car. *Social Inclusion*, *2* (4), 1-13. https://doi.org/10.17645/si.v2i4.127

Crenshaw, K. (1991). Race, gender, and sexual harassment. *Southern California Law Review, 65*, 1467-1496. https://doi.org/10.2139/ssrn.1434678

Currie, G., Richardson, T., Smyth, P., Vella-Brodrick, D., Hine, J., Lucas, K., ... & Stanley, J. (2010). Investigating links between transport disadvantage, social exclusion and well-being in Melbourne–Updated results. *Research in transportation economics*, *29*(1), 287-295. https://doi.org/10.1016/j.retrec.2010.07.036

D'Agostino, C., Piva, E., Pucci, P. & Rossi, C. (2024). A systematic literature review on women's daily mobility in the Global North. *Transport Reviews*, *44* (5), 1016-1044. http://dx.doi.org/10.1080/01441647.2024.2350613

D'Amico A. (2024). Strategies and instruments for active mobility: the main Italian policies. *TeMA - Journal of Land Use, Mobility and Environment, 17*(3), 509-515. https://doi.org/10.6093/1970-9870/11166

Dennis, C., Alamanos, E., Papagiannidis, S. & Bourlakis, M. (2016). Does social exclusion influence multiple channel use? The interconnections with community, happiness, and well-being. *Journal of Business Research, 69* (3), 1061-1070. https://doi.org/10.1016/j.jbusres.2015.08.019

Dharmowijoyo, D. B. E. & Joewono, T. B. (2019). Mobility and health: The interaction of activity-travel patterns, overall well-being, transport-related social exclusion on health parameters. In *Energy Efficiency in Mobility Systems*, 53-83. https://doi.org/10.1007/978-981-15-0102-9\_4

Di Ciommo, F. & Shiftan, Y. (2017). Transport equity analysis. *Transport Reviews, 37* (2), 139-151. https://doi.org/ 10.1080/01441647.2017.1278647

Di Ruocco, I. (2022a). Resilient marginal cities by encouraging intermodality strategies. *TeMA - Journal of Land Use, Mobility and Environment, 15* (3), 377-396. https://doi.org/10.6092/1970-9870/9031

Di Ruocco, I. (2022b). Ripensare gli spazi urbani con l'integrazione di politiche di MaaS. Un incubatore sociale per promuovere la mobilità dei soggetti fragili. In *Dare valore ai valori in urbanistica – Worthing values for urban planning, XXIV Conference 2022*, Brescia, 23rd-24th June 2022. Planum Publisher.

Di Ruocco, I. (2024). Eco-mobility justice in the ecological transition. An analysis for possible directions in mobility and transport equity. *TeMA - Journal of Land Use, Mobility and Environment (2),* 97-111. https://doi.org/10.6093/1970-9870/10162

Dixon, S., Irshad, H., Pankratz, D. M. & Bornstein, J. (2019). *The 2019 Deloitte City Mobility Index: Gauging Global Readiness for the Future of Mobility*. Deloitte Insights. Retrieved from: https://www2.deloitte.com/content/dam/insights /us/articles/4331\_Deloitte-City-Mobility-Index/4331\_city-mobility-index\_OVERVIEW.pdf

Egger, S. M., Gemperli, A., Filippo, M., Liechti, R. & Gantschnig, B. E. (2022). The experiences and needs of persons with disabilities in using paratransit services. *Disability and Health Journal, 15* (4), 101365. https://doi.org/10.1016/j.dhjo.2022.101365

EIT Urban Mobility. (2021). *Full report: Urban mobility strategies during COVID-19.* Retrieved from https://www.eiturbanmobility.eu/wp-content/uploads/2021/03/Urban-mobility-strategies-during-COVID-19\_long-1.pdf

El-Geneidy, A., Levinson, D., Diab, E., Boisjoly, G., Verbich, D. & Loong, C. (2016). The cost of equity: Assessing transit accessibility and social disparity using total travel cost. *Transportation Research Part A: Policy and Practice, 91*, 302-316. https://doi.org/10.1016/j.tra.2016.07.003

European Union: EU (2014). Commission Regulation (EU) No 1300/2014 of 18 November 2014 on the technical specifications for interoperability relating to accessibility of the Union's rail system for persons with disabilities and persons with reduced mobility. Retrieved from https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:32014R1300

European Union: EU (2019). *Directive (EU) 2019/882 of the European Parliament and of the Council of 17 April 2019 on the accessibility requirements for products and services.* Retrieved from https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:32019L0882

*Eurostat (2022). Living conditions in Europe - poverty and social exclusion.* Retrieved from https://ec.europa.eu/eurostat/ data/database

Ferrari, L., Berlingerio, M., Calabrese, F. & Reades, J. (2014). Improving the accessibility of urban transportation networks for people with disabilities. *Transportation Research Part C: Emerging Technologies, 45*, 27-40. https://doi.org/10.1016/j.trc.2013.10.005

Gargiulo, C., Zucaro, F. & Gaglione, F. (2018). A set of variables for the elderly accessibility in urban areas. *TeMA - Journal of Land Use, Mobility and Environment*, 53-66. https://doi.org/10.6092/1970-9870/5738

Gatrell, A. C. (2013). Therapeutic mobilities: Walking and 'steps' to wellbeing and health. *Health & Place, 22*, 98-106. https://doi.org/10.1016/j.healthplace.2013.04.002

Guida, C., Angiello, G., Tulisi, A., Gaglione, F. & Morosini, R. (2020). Reviews pages: The city challenges and external agents. Methods, tools and best practices. *TeMA - Journal of Land Use, Mobility and Environment, 13* (1), 99-119. https://doi.org/10.6092/1970-9870/6793 Guzman, L. A., Oviedo, D. & Rivera, C. (2017). Assessing equity in transport accessibility to work and study: The Bogotá region. *Journal of Transport Geography, 58*, 236-246. http://dx.doi.org/10.1016/j.jtrangeo.2016.12.016

Guzman, L. A. & Oviedo, D. (2018). Accessibility, affordability and equity: Assessing 'pro-poor' public transport subsidies in Bogotá. *Transport Policy, 68*, 37-51. https://doi.org/10.1016/j.tranpol.2018.04.012

Hananel, R. & Berechman, J. (2016). Justice and transportation decision-making: The capabilities approach. *Transport Policy*, 49, 78-85. https://doi.org/10.1016/j.tranpol.2016.04.005

He, S. Y., Thøgersen, J., Cheung, Y. H. Y. & Yu, A. H. Y. (2020). Ageing in a transit-oriented city: Satisfaction with transport, social inclusion and wellbeing. *Transport Policy*, *97*, 85-94. https://doi.org/10.1016/j.tranpol.2020.06.016

Hezam, I. M., Mishra, A. R., Rani, P. & Alshamrani, A. (2022). Assessing the barriers of digitally sustainable transportation system for persons with disabilities using Fermatean fuzzy double normalization-based multiple aggregation method. *Applied Soft Computing*, *133*, 109910. http://dx.doi.org/10.1016/j.asoc.2022.109910

Hidayati, I., Yamu, C. & Tan, W. (2019). The emergence of mobility inequality in greater Jakarta, Indonesia: A socio-spatial analysis of path dependencies in transport–land use policies. *Sustainability*, *11* (18), 5115. https://doi.org/10.3390/su11185115

Hine, J. & Mitchell, F. (2016). *Transport disadvantage and social exclusion: Exclusionary mechanisms in transport in urban Scotland.* Routledge.

Hranický, M. P., Nedeliaková, E. & Kuka, A. (2021). Innovations in the process of transporting the disabled. *Transportation Research Procedia*, *55*, 450-457. http://dx.doi.org/10.1016/j.trpro.2021.07.008

Ignaccolo, M., Inturri, G., Giuffrida, N. & Torrisi, V. (2016). Public transport accessibility and social exclusion: Making the connections. In *Proceedings of the Third International Conference on Traffic and Transport Engineering (ICTTE),* Belgrade, Serbia, 24-25. https://www.researchgate.net/publication/311589366\_PUBLIC\_TRANSPORT\_ACCESSIBILITY\_AND\_SOCIAL \_EXCLUSION\_MAKING\_THE\_CONNECTIONS

Kuriakose, B., Shrestha, R. & Sandnes, F. E. (2020). Smartphone navigation support for blind and visually impaired people: A comprehensive analysis of potentials and opportunities. In *Universal Access in Human-Computer Interaction. Applications and Practice: 14th International Conference, UAHCI 2020, Held as Part of the 22nd HCI International Conference, HCII 2020, Copenhagen, Denmark, July 19-24, 2020, Proceedings, Part II, 568-583. Springer International Publishing. https://doi.org/10.1007/978-3-030-49108-6\_41* 

Kuttler, T. & Moraglio, M. (Eds.). (2023). *Re-thinking mobility poverty: Understanding users' geographies, backgrounds and aptitudes.* Routledge.

Kwakye, K. (2022). Social responsibility considerations for autonomous vehicles implementation: Design and legislature for equity towards vulnerable groups. *SSRN Electronic Journal*. https://dx.doi.org/10.2139/ssrn.4175764

Jones, M., Morris, J. & Deruyter, F. (2018). Mobile healthcare and people with disabilities: current state and future needs. *International journal of environmental research and public health*, *15* (3), 515. https://doi.org/ 10.3390/ijerph15030515

LaPlante, M. P. (2014). Key goals and indicators for successful aging of adults with early-onset disability. *Disability and Health Journal*, 7(1), S44-S50. https://doi.org/10.1016/j.dhjo.2013.08.005

Lee, Y., Oh, S., Cha, J., Kim, K., Lee, S. Y. & Kim, H. (2022). Mobility and life satisfaction among wheelchair users: A study using mobile phone data in Seoul, South Korea. *International Journal of Social Welfare, 31* (3), 368-381. https://doi.org/10.1111/ijsw.12529

Li, L. & Loo, B. P. Y. (2017). Mobility impairment, social engagement, and life satisfaction among the older population in China: A structural equation modeling analysis. *Quality of Life Research, 26*, 1273-1282. https://doi.org/10.1007/s11136-016-1444-x

Litman, T. (2022). *Evaluating transportation equity: Guidance for incorporating distributional impacts in transportation planning.* Victoria Transport Policy Institute. https://vtpi.org/Litman\_ITEJ\_Equity\_Apr2022.pdf

Lucas, K. (2011). Making the connections between transport disadvantage and the social exclusion of low-income populations in the Tshwane Region of South Africa. *Journal of Transport Geography, 19* (6), 1320-1334. https://doi.org/ 10.1016/j.jtrangeo.2011.02.007

Lucas, K. (2012). Transport and social exclusion: Where are we now? *Transport Policy, 20*, 105-113. https://doi.org/10 .1016/j.tranpol.2012.01.013

Lucas, K., Mattioli, G., Verlinghieri, E. & Guzman, A. (2016). Transport poverty and its adverse social consequences. In *Proceedings of the Institution of Civil Engineers-Transport, 169* (6), 353-365. Thomas Telford Ltd. https://doi.org/10.1680/jtran.15.00073

Ma, L., Kent, J. & Mulley, C. (2018). Transport disadvantage, social exclusion, and subjective well-being: The role of the neighborhood environment—evidence from Sydney, Australia. *Journal of Transport and Land Use, 11* (1), 31-47. https://doi.org/10.5198/jtlu.2018.1008

Martens, K. (2019). Why accessibility measurement is not merely an option, but an absolute necessity. In *Designing Accessibility Instruments* (pp. 37-51). Routledge. http://dx.doi.org/10.4324/9781315463612-4

Martinelli V. (2024). Governance of the energy transition: the role of local authorities in Italy. *TeMA - Journal of Land Use, Mobility and Environment,* 17(3), 503-507. https://doi.org/10.6093/1970-9870/11168

Mattsson, L. & Jenelius, E. (2015). Vulnerability and resilience of transport systems–A discussion of recent research. *Transportation Research Part A: Policy and Practice, 81*, 16-34. https://doi.org/10.1016/j.tra.2015.06.002

Milbourne, P. & Kitchen, L. (2014). Rural mobilities: Connecting movement and fixity in rural places. *Journal of Rural Studies, 34*, 326-336. https://doi.org/10.1016/j.jrurstud.2014.01.004

Moore, J., Lucas, K. & Bates, J. (2013). *Social disadvantage and transport in the UK: A trip-based approach*. Retrieved from https://www.tsu.ox.ac.uk/pubs/1063-moore-lucas-bates.pdf

Mora, H., Gilart-Iglesias, V., Pérez-del Hoyo, R. & Andújar-Montoya, M. D. (2017). A comprehensive system for monitoring urban accessibility in smart cities. *Sensors (Basel, Switzerland), 17*(8), 1834. https://doi.org/10.3390/s17081834

Muller, M., Park, S., Lee, R., Fusco, B. & Correia, G. H. d. A. (2021). Review of whole system simulation methodologies for assessing mobility as a service (MaaS) as an enabler for sustainable urban mobility. *Sustainability*, *13* (10), 5591. https://doi.org/10.3390/su13105591

Niehaus, M., Galilea, P. & Hurtubia, R. (2016). Accessibility and equity: An approach for wider transport project assessment in Chile. *Research in Transportation Economics*, *59*, 412-422. https://doi.org/10.1016/j.retrec.2016.05.003

Nixon, D. V. & Schwanen, T. (2018). Emergent and integrated justice: Lessons from community initiatives to improve infrastructures for walking and cycling. In N. Cook & D. Butz (Eds.), *Mobilities, Mobility Justice and Social Justice*, 129-141. Routledge. http://dx.doi.org/10.4324/9780815377047

Nutley, S. (2003). *Bridging the policy/research divide: Reflections and lessons from the UK*. Research Unit for Research Utilisation. https://www.treasury.govt.nz/sites/default/files/2007-09/tgls-nutley.pdf

O'Neill, Y. & O'Mahony, M. (2005). Travel behavior and transportation needs of people with disabilities: Case study of some categories of disability in Dublin, Ireland. *Transportation Research Record: Journal of the Transportation Research Board, 1924*, 1-8. https://doi.org/10.3141/1924-01

OguzS. & TanyasM. (2024). Technological applications in sustainable urban logistics: a systematic review with bibliometric analysis. *TeMA - Journal of Land Use, Mobility and Environment, 17*(3), 487-502. https://doi.org/10.6093/1970-9870/10844

Oviedo, D. & Sabogal, O. (2020). Unpacking the connections between transport and well-being in socially disadvantaged communities: Structural equations approach to low-income neighbourhoods in Nigeria. *Journal of Transport & Health, 19,* 100966. https://doi.org/10.1016/j.jth.2020.100966

Paiva, S., Amaral, A., Gonçalves, J., Lima, R. & Barreto, L. (2022). Image recognition-based architecture to enhance inclusive mobility of visually impaired people in smart and urban environments. *Sustainability, 14* (18), 11567. https://doi.org/ 10.3390/su141811567

Pan, Y. & He, S. Y. (2022). Analyzing COVID-19's impact on the travel mobility of various social groups in China's Greater Bay Area via mobile phone big data. *Transportation Research Part A: Policy and Practice, 159*, 263-281. https://doi.org/10.1016/j.tra.2022.03.015

Papa, E., Carpentieri, G. & Guida, C. (2018). Measuring walking accessibility to public transport for the elderly: The case of Naples. *TeMA - Journal of Land Use, Mobility and Environment,* 105-116. https://doi.org/10.6092/1970-9870/5766

Papagiannidis, S., Bourlakis, M., Alamanos, E. & Dennis, C. (2017). Preferences of smart shopping channels and their impact on perceived well-being and social inclusion. *Computers in Human Behavior, 77*, 396-405. https://doi.org/10.1016 /j.chb.2017.04.029

Park, J. & Chowdhury, S. (2018). Investigating the barriers in a typical journey by public transport users with disabilities. *Journal of Transport & Health, 10*, 361-368. https://doi.org/10.1016/j.jth.2018.05.008

Park, K., Esfahani, H. N., Novack, V. L., Sheen, J., Hadayeghi, H., Song, Z. & Christensen, K. (2022). Impacts of disability on daily travel behaviour: A systematic review. *Transport Reviews*, *43* (2), 178-203. https://doi.org/10.1080/01441647.2022.2060371

Pezzagno, M. & Richiedei, A. (2022). New scenarios for safe mobility in urban areas: Emerging topics from an international debate. *TeMA - Journal of Land Use, Mobility and Environment,* 243-251. https://doi.org/10.6093/1970-9870/8649

POLIS & Rupprecht Consult - Forschung & Beratung GmbH (Eds.) (2021). *Topic guide: Planning for more resilient and robust urban mobility.* Retrieved from https://civitas.eu/sites/default/files/sump\_topic-guide\_planning\_for\_more\_resilient\_and\_robust\_urban\_mobility.pdf

Preston, J. & Rajé, F. (2007). Accessibility, mobility and transport-related social exclusion. *Journal of Transport Geography*, 15 (3), 151-160. https://doi.org/10.1016/j.jtrangeo.2006.05.002

Pretty, G., Rapley, M. & Bramston, P. (2002). Neighbourhood and community experience, and the quality of life of rural adolescents with and without an intellectual disability. *Journal of Intellectual and Developmental Disability*, *27*(2), 106-116. https://doi.org/10.1080/13668250220135079-5 Pucci, P., Vecchio, G., Bocchimuzzi, L. & Lanza, G. (2019). Inequalities in job-related accessibility: Testing an evaluative approach and its policy relevance in Buenos Aires. *Applied Geography, 107*, 1-11. https://doi.org/10.1016/j.apgeog.2019.04.002

Remillard, E. T., Campbell, M. L., Koon, L. M. & Rogers, W. A. (2022). Transportation challenges for persons aging with mobility disability: Qualitative insights and policy implications. *Disability and Health Journal, 15*(1), 101209. https://doi.org/10.1016/j.dhjo.2021.101209

Riazi, A., Riazi, F., Yoosfi, R. & Bahmeei, F. (2016). Outdoor difficulties experienced by a group of visually impaired Iranian people. *Journal of Current Ophthalmology, 28 (2)*, 85–90. https://doi.org/10.1016/j.joco.2016.04.002

Ryan, J. & Pereira, R. H. (2021). What are we missing when we measure accessibility? Comparing calculated and self-reported accounts among older people. *Journal of Transport Geography, 93* (4), 103086. https://doi.org/10.1016 /j.jtrangeo.2021.103086

Sabella, S. A. & Bezyak, J. L. (2019). Barriers to public transportation and employment: A national survey of individuals with disabilities. *Journal of Applied Rehabilitation Counseling*, *50* (3), 174–185. https://doi.org/10.1891/0047-2220.50.3.174

Sen, S., Charles, M. B. & Harrison, J. L. (2022). Usage-based road pricing and potential equity issues: A study of commuters in South East Queensland, Australia. *Transport Policy, 118*, 33-43. https://doi.org/10.1016/j.tranpol.2022.01.006

Shareck, M., Frohlich, K. L. & Kestens, Y. (2014). Considering daily mobility for a more comprehensive understanding of contextual effects on social inequalities in health: A conceptual proposal. *Health & Place, 29*, 154-160. https://doi.org/10.1016/j.healthplace.2014.07.007

Sheller, M. (2018). Mobility justice: The politics of movement in an age of extremes. Verso Books.

Shergold, I. & Parkhurst, G. (2012). Transport-related social exclusion amongst older people in rural Southwest England and Wales. *Journal of Rural Studies, 28* (4), 412–421. https://doi.org/10.1016/j.jrurstud.2012.01.010

Shliselberg, R. & Givoni, M. (2017). Motility as a policy objective. *Transport Reviews, 38* (3), 279-297. https://doi.org/10.1080/01441647.2017.1355855

Sochor, J. (2014). The reassuring effects of ICT in public transportation: The perspectives of older adults. *Proceedings of* the 93rd Annual Meeting of the Transportation Research Board, Washington, DC, USA.

Sochor, J. (2015). Enhancing mobility and perceived safety via ICT: The case of a navigation system for visually impaired users. In V. Ceccato & A. Newton, (Eds.), *Safety and Security in Transit Environments: An Interdisciplinary Approach* (pp. 344-361). Springer Nature. https://doi.org/10.1057/9781137457653\_19

Soja, E. W. (2013). Seeking spatial justice, 16. University of Minnesota Press.

Soltani, S. H. K., Sham, M., Awang, M. & Yaman, R. (2012). Accessibility for disabled in public transportation terminal. *Procedia - Social and Behavioral Sciences, 35*, 89-96. https://doi.org/10.1016/j.sbspro.2012.02.066

Stanley, J. K., Hensher, D. A., Stanley, J. R. & Vella-Brodrick, D. (2011). Mobility, social exclusion and well-being: Exploring the links. *Transportation Research Part A: Policy and Practice*, 45 (8), 789-801. http://dx.doi.org/10.1016/j.tra.2011.06.007

Stanley, J., Stanley, J. & Hensher, D. (2012). Mobility, social capital and sense of community: What value? *Urban Studies,* 49 (16), 3595-3609. http://dx.doi.org/10.1177/0042098012447002

Stanley, J. & Stanley, J. (2017). The importance of transport for social inclusion. *Social Inclusion, 5* (4), 108-115. http://dx.doi.org/10.17645/si.v5i4.1289

Stanley, J., Stanley, J., Balbontin, C. & Hensher, D. (2019). Social exclusion: The roles of mobility and bridging social capital in regional Australia. *Transportation Research Part A: Policy and Practice, 125*, 223-233. https://doi.org/10.1016 /j.tra.2018.05.015

Stanley, J. K., Hensher, D. A. & Stanley, J. R. (2022). Place-based disadvantage, social exclusion and the value of mobility. *Transportation Research Part A: Policy and Practice, 160,* 101-113. https://doi.org/10.1016/j.tra.2022.04.005

Titheridge, H., Christie, N., Mackett, R., Oviedo Hernández, D. & Ye, R. (2014). *Transport and poverty: A review of the evidence*. Retrieved from https://www.ucl.ac.uk/transport/sites/transport/files/transport-poverty.pdf

Uteng, T. P. & Cresswell, T. (Eds.). (2008). Gendered mobilities (p. 270). Aldershot: Ashgate.

van Dülmen, C., Šimon, M. & Klärner, A. (2022). Transport poverty meets car dependency: A GPS tracking study of socially disadvantaged groups in European rural peripheries. *Journal of Transport Geography, 101*, 103351. https://doi.org/ 10.1016/j.jtrangeo.2022.103351

Van Wee, B. & Geurs, K. (2011). Discussing equity and social exclusion in accessibility evaluations. *European journal of transport and infrastructure research*, *11* (4). https://doi.org/10.18757/ejtir.2011.11.4.2940

Vecchio, G. & Martens, K. (2021). Accessibility and the capabilities approach: A review of the literature and proposal for conceptual advancements. *Transport Reviews*, *41* (6), 833–854. https://doi.org/10.1080/01441647.2021.1931551

Velho, R. (2019). Transport accessibility for wheelchair users: A qualitative analysis of inclusion and health. *International Journal of Transportation Science and Technology*, 8 (2), 103–115. https://doi.org/10.1016/j.ijtst.2018.04.005

Vitale Brovarone, E. (2021). Accessibility and mobility in peripheral areas: A national place-based policy. *European Planning Studies*, *30*(8), 1444-1463. https://doi.org/10.1080/09654313.2021.1894098

Wan, C., Yang, Z., Zhang, D., Yan, X. & Fan, S. (2018). Resilience in transportation systems: A systematic review and future directions. *Transport Reviews*, *38* (4), 479-498. https://doi.org/10.1080/01441647.2017.1383532

Welch, T. F. & Mishra, S. (2013). A measure of equity for public transit connectivity. *Journal of Transport Geography, 33*, 29-41. https://doi.org/10.1016/j.jtrangeo.2013.09.007

Wong, S. (2018). Traveling with blindness: A qualitative space-time approach to understanding visual impairment and urban mobility. *Health & Place, 49*, 85-92. https://doi.org/10.1016/j.healthplace.2017.11.009

World Health Organization: WHO (2023). *Blindness and vision impairment*. Retrieved from https://www.who.int/news-room/fact-sheets/detail/blindness-and-visual-impairment

Xia, J. C., Nesbitt, J., Daley, R., Najnin, A., Litman, T. & Tiwari, S. P. (2016). A multi-dimensional view of transport-related social exclusion: A comparative study of Greater Perth and Sydney. *Transportation Research Part A: Policy and Practice, 94*, 205-221. https://doi.org/10.1016/j.tra.2016.09.009

Zhou, H., Hou, K., Zuo, D. & Li, J. (2012). Intelligent urban public transportation for accessibility dedicated to people with disabilities. *Sensors*, *12* (8), 10678-10692. https://doi.org/10.3390/s120810678

#### Author's profile

#### Irina Di Ruocco

She is a transport engineer with a degree from the University of Naples Federico II, is currently a PhD student in Methods and Models for Economic Decisions (MMED) at the University of Insubria. Her expertise includes sustainable mobility, infrastructure project evaluation, economic analysis, and European planning, particularly within EU projects. With experience in urban planning and transportation initiatives like waterfronts, ports, cycleways, and car-sharing, she has also contributed to several scientific publications in the field.